Author's response to reviews

Title: Statins and risk of gastric cancer in diabetes patients

Authors:

Jeeyun Lee (jyunlee@skku.edu)
Soo Hyeon Lee (taro6334@naver.com)
Kyu Yeon Hur (taro111@naver.com)
Sook Young Woo (sookyoung.woo@samsung.com)
Sun Woo Kim (sunwoo.kim@samsung.com)
Won Ki Kang (wkkang@skku.edu)

Version: 5 Date: 12 November 2012

Author's response to reviews:

As recommneded by the editor, we have added the following statements:

Competing Interests: The authors declare that they have no competing interests.
Authors’ contributions: JL, SHL drafted the manuscript. KHR, SYO, SWK participated in the design of the study and performed the statistical analysis. WK conceived of the study and participated in its design and coordination. All authors read and approved the final manuscript.